\begin{table}%t1
\par
\caption{\label{tab:bursts}Comparison of peak fluxes and luminosities  of the AXP bursts.}
%\centerline
{\begin{tabular}{llll}
\hline\hline
Name & $F$& $L$& Reference\\ & [$10^{-8}$~erg~cm$^{-2}$s$^{-1}$] & [erg~s$^{-1}$] & \\
\hline\noalign{\smallskip}
4U 0142+61 & $0.3\div 4$~(2--30~keV)&$\le$$1.7\times 10^{40}$&1\\
 1E 1048-59 &  $0.5\div 1.1$~(2--40~keV)&$\le$$1.6\times 10^{37}$&2\\
\axp& $>$$3\times 10^4$~($>$$80$~keV)&$>$$2.7\times 10^{42}$&this work\\
CXOU J1647-45 & $\simeq$$1$~(15--150~keV)&$\le$$1.7\times 10^{37}$&4\\
XTE J1810-197 & $0.07\div 10$~(2--30~keV)&$\le$$10^{38}$&5\\
1E 2259+586 & $0.1\div 40$~(2--60~keV)&$\le$$2.5 \times 10^{39}$&6\\ 
\hline
\end{tabular}}
\par
\tablebib{{(1)} \citet{gavriil07}; (2) \citet{gavriil02}; (3) this work; (4) \citet{israel07}; (5) \citet{woods05};  (6) \citet{kaspi03,gavriil04}.}
\end{table}